Modeling Large-Scale Manufacturing of Lithium-Ion Battery Cells: Impact of New Technologies on Production Economics

Florian Degen^(D) and Oliver Krätzig

Abstract—The global demand for electric vehicles is increasing exponentially, as is the demand for lithium-ion battery cells. This has led to a strong ongoing competition among companies to achieve the lowest battery cell production cost. Herein, to provide guidance on the identification of the best starting points to reduce production costs, a bottom-up cost calculation technique, process-based cost modeling (PBCM), for battery cell production is reproduced and validated by drawing on a consistent dataset of a real battery cell production plant. The model is based on teardowns of a real battery cell factory and will prove useful for planning activities of today's, so-called, "giga factories." The PBCM performed in the present study involves discussions on, e.g., production balancing, relocation of factories to low-wage countries, usage of new production and cell technologies, etc. The use of novel approaches, such as tabless cell design, dry coating, and NMC811 chemistry, is discussed. Finally, the ways in which battery cell production costs can be reduced further in the forthcoming years are shown, and implications for researchers, practitioners, and policy makers are provided.

Index Terms—Battery cell, cost optimization, process-based cost modeling (PBCM), production costs.

I. INTRODUCTION

T HE worldwide demand for battery cells is increasing exponentially, driven particularly by the growing success of electric vehicles (EVs). Current studies predict a demand of approximately 2600 GWh/a for 2030 [1]. Others even predict demands of up to 10 000 GWh/a, but without referencing to an exact year [2]. As a result, battery cell production capacity is being rapidly expanded worldwide; e.g., by the end of 2020, 800 GWh of battery cell production capacity was announced or planned in Europe alone [3]. A change in this trend is not expected at present. Such a scenario has led to a strong ongoing price competition among battery manufacturers. The literature reports on different effects that contribute to the cost reduction

Manuscript received 14 February 2021; revised 4 December 2021, 22 March 2022, and 28 July 2022; accepted 23 March 2023. This work was supported by the German Federal Ministry of Education and Research under Grant 03XP0256. Review of this manuscript was arranged by Department Editor Y. Zhou. (*Corresponding author: Florian Degen.*)

The authors are with the Fraunhofer Research Institution for Battery Cell Production FFB, 48165 Münster, Germany (e-mail: florian.degen@ffb.fraunhofer.de; oliver.kraetzig@ffb.fraunhofer.de).

Color versions of one or more figures in this article are available at https://doi.org/10.1109/TEM.2023.3264294.

Digital Object Identifier 10.1109/TEM.2023.3264294

of battery technology. For example, it has been reported that economies of scale-cost reduction through learning rate after increasing production quantity (in this case doubling)-can reduce the price as much as 6%-9% per battery pack [4]. The development of optimized materials for use in lithium-ion cells, such as active materials for cathodes, and electrolytes, as well as corresponding inactive materials, can help achieve higher energy densities and lifetimes, thereby indirectly reducing costs at the cell level since less material is required per cell [5]. In addition, price advantages in terms of more efficient methods of material synthesis can be passed on to cell-producing customers. Another crucial cost-reducing factor is the optimization of the battery cell production process. Studies have reported high cost sensitivity of electrode production, advanced formation and aging processes, continuous process control, and process substitution as targets for innovative process technologies [6]. The integration of new production technologies is required to reduce prices even further [2]. However, for many production researchers and manufacturing engineers, it is rather difficult to identify the part of the process chain for which new production technologies are needed the most and the economic impact these new approaches might bring. One major reason is that the battery cell cost and, in particular, the exact cost structure is somewhat complex and involves variables other than those from usual cell and process design activities. The general obstacles that inhibit production of reliable battery cells that are cost efficient are as follows.

- 1) Costs usually depend on industry activity and are, therefore, strongly **confidential**.
- The battery market is not fully developed yet. Battery cells, especially for EVs, are being produced in significant amounts only for the last couple of years. Thus, the current knowledge and experience base for EV applications is still inadequate.
- As the market is relatively young and growing strongly, production know-how is being learned continuously, leading to a decrease in production costs.
- 4) There is no stable ratio between supply and demand. Both are changing rapidly and have a substantial impact on costs and prices. Other influencing factors include market structures such as possible oligopolies and lack of information availability, the discovery of new raw material sources and development of efficient refining methods,

and the use of new cell chemistries (which changes the demand for existing materials and thus costs). Note that the terms "price" and "cost" are often used simultaneously in existing studies.

5) Battery cell costs **depend** strongly on material prices, and which in turn depend significantly on the **quantity of purchase**, which varies as well.

As every business-oriented organization aims to maximize profit while reducing costs under the given conditions, the same is true for battery cell production. A literature analysis by Duffner et al. [7] from 2020 shows an average cost decrease of approximately 50% for lithium-ion battery cells between 2015 and 2020. More importantly, even today, the cost assumptions range from 84 [8] to 138 €/kWh [3] for state-of-the-art lithium-ion NMC cell chemistry. Thus, there is a large variation in reported costs, in general and over time. In addition, many costs for battery cells are not split in terms of material and production costs. There are studies that try to analyze the cost structure, but strong differences can be observed here as well. The German Engineering Association (VDMA) reports, for example, material cost proportions of 60% [3], while BMW has reported material cost proportions of 80% [9]. Therefore, there are large differences in quotable battery cell production costs, and it is difficult to elaborate on their validity and how they are calculated. Similarly, there are papers and publications that explain the breakup of production costs through different methodological approaches (e.g., [3], [7], [10], [11], [12], [13]). Most of these studies gathered information from various other sources (secondary data). However, the combination of these data from different sources for a cost model is not perfect because of the following.

- 1) The primary source and its credibility are mostly unknown.
- 2) The locations of the analyzed production sites differ, which has a significant effect on production costs.
- The exact time of data collection is unknown. Owing to the rapid developments in battery cell production, the life of available information diminishes fast.
- 4) The exact cell chemistry is unknown or not named, even though it has a significant effect on cost per kWh.
- 5) The cell format is unknown, which also has a significant effect on the cost per kWh.
- 6) The cell design is unknown. If a cell design has a high-power layout or a high-energy layout, it also has a significant effect on the cost per kWh.

Despite the abovementioned shortcomings, a cost model involving real production-related large-scale data on state-of-theart battery production technology, allowing for full transparency and access to primary data, is missing. In addition, the effects of new battery cell manufacturing technologies on production economics are largely unknown.

The present study aims to contribute toward answering the following question by considering battery cell production as a specific case: How can the manufacturing costs be improved by new process technologies or general improvement measures?

This study is settled in the field of technology assessment and offers implications for literature on production economics and technology planning. Our work contributes in particular to reducing complexity of technology assessment problems in the light of integrating new and emerging technologies. This is achieved by methodically disclosing the cost structure of battery cell production (as exemplary processing industry) in a generalizable way and elaborating on the impact, which results from the integration of new and emerging technologies. It furthermore adds to the robustness of bottom-up cost modeling by providing validation for the case of real battery cell fabrication. Implications for technology managers, researchers, and policy responses are discussed as well.

The rest of this article is organized as follows. Section II introduces the research work of the present study by first referring to the literature on production economics. Subsequently, existing preliminary work on cost modeling of battery cell production is recapitulated, and special features of the field of battery research are highlighted. This emphasizes the relevance of the topic from the perspective of the academic discipline of innovation and technology management. Section III outlines the case study of a lithium-ion battery cell manufacturing facility analyzed in this study by first discussing its specific background (funding and task profile). Subsequently, actual planning data on the facility layout, the production throughput, cost-relevant equipment specifications, and the production balancing are provided. Section IV describes the chosen methodological approach for cost modeling and discusses it with alternative approaches. Section V provides an overview of production costs and discusses the influence of production-related optimization factors. Subsequently, the influence of technological innovations on production is described, and an outlook on recommendations for action derived from these is given. Finally, Section VI concludes the article with the contributions and limitations as well as recommendations for future research.

II. LITERATURE REVIEW

Meeting cost targets is, besides achieving quality and throughput targets, arguably an essential component of decisions on the use of technologies in the production process. Work in the scientific field of engineering and technology management, which is closely related to the subject of this study, focuses on the development and testing of methodology for problems of technology assessment [14] and decision-making [15], [16]. Although the studies discussed in the following originate from multidisciplinary scientific streams, they provide a solid overview of cost structures in battery cell production as an exemplary manufacturing sector, which is a basis for understanding possible cost-reducing measures in the context of technology assessment.

Even though recent studies have reported annual cost reductions of approximately 8% (between 2007 and 2014) [4] and promising cost levels of \$70–90/kWh for cells and \$150/kWh per pack [17], [18], there are a variety of variables that can lead to the realization of these cost reduction potentials. Different strategies exist to achieve these goals, which have been discussed and investigated in various publications. Thus, in terms of the cost structures of lithium-ion battery cells, it is well known that a large part is due to the sourcing and further processing of battery materials (especially cathode, anode, and electrolyte). Certain studies even see material price as the central limiting factor for the price reduction potential of lithium-ion batteries [12]. Purchasing large quantities to reduce costs per unit quantity are suggested as strategies to achieve cost targets. In addition, inhouse production of materials or precursors is under discussion for backward integration along the value chain. This helps extend value creation processing within the company [19], which also involves the establishment of strategic partnerships [6]. Certain researchers have addressed their cost forecasts on battery cells and packs accordingly, examining the impact of battery materials utilized [5], [20], [21]

In addition to the strategic approaches to input factors aimed at increasing cost efficiency, there is a wide scope for actions required to realize cost degression effects. These concern the actual production process, for example, by balancing the production size. Studies that address the size of production scale define a corridor from 200 to 300 MWh/year to 2 GWh/year as the most efficient so far [10], [22]. With the prospect of positive effects of balancing (in this case, the capacitive load of electrode production), larger factories can also be operated in a cost-optimized manner in the future. This effect was quantified as a cost advantage of >\$5 kWh-1 [10]. Studies that address the impact of compound effects assume savings of between 9% and 21% per battery pack based on bottom-up calculations [11]. In addition to the design of the size scaling of production processes, the production location, and associated costs due to employee wages, energy prices, building prices, and tax levies are other key issues that influence production costs. Studies attribute a cost difference of \$6.4/kWh to the location decision and address factors indirectly affecting production, such as GHG emissions through energy mix, knowledge structure, labor market, and industrial development [23], [24], [25]. Closely related to the approach of using business strategies to reduce material costs is the approach of fostering technological efficiency to optimize resource allocation. For example, studies address critical resources and their alternatives [26] on the one hand, and material recycling opportunities or reusability concepts [27], [28] on the other. Optimization work with the aim of reducing energy consumption [29], as well as elaboration on the digitalization of the production environment and quality assurance have so far been primarily simulation based [30], [31], [32], [33], [34]. The establishment of continuous process control enables a more uniform design of manufacturing stages, potentially leading to lower buffer inventories, and thus, enabling a lean, cost-optimized manufacturing environment [7], [35].

Another way that is widely discussed in academic literature is the integration of technological innovations to substitute or streamline the required inputs. For example, it has been shown that investments in plant equipment show potential savings by eliminating solvent recovery equipment (when processing with NMP). This is enabled by alternative processing routes using aqueous solvents [36]. The use of dry coating processes that eliminate solvents can lead to substantial savings in terms of equipment investment and ongoing costs due to energy consumption [37].

The cell finishing, the process section at the end of the manufacturing chain for lithium-ion battery cells, comprises the

steps of formation and cell aging. The formation step refers to the formation of the solid electrolyte interface (SEI), an interface layer at the electrodes of the battery cell that allows physical and chemical thermodynamic processes to take place by applying voltage and running a time- and phase-dependent program [38], [39]. The formation step is particularly quality determining and has an enormous impact on cell performance [6], [40]. However, formation is considered one of the most cost-intensive processes in battery cell manufacturing due to its enormous energy consumption. Therefore, cost reduction strategies often deal with reducing the time of the forming cycles [41]. This holds implications for the energy costs required and involves forming strategies to accelerate the formation of the necessary boundary layer between the electrode and electrolyte [42], [43]. The following examples, among others, are known in the literature as strategies to save time.

- 1) Narrowing the voltage window of the formation cycle can effectively shorten the formation time [40].
- 2) The application of pulsed current charging enables a higher charging rate in the formation process, and thus, shortens the formation time [44].
- The application of simultaneous elevated temperature and mechanical loading resulted in time reductions in the formation process [41].
- 4) The approach of applying an artificial SEI layer by thin film techniques such as atomic layer deposition, rather than its formation out of the components of the cell (primarily the electrolyte) by formation programs, has so far only been demonstrated on a laboratory scale [45], [46], [47], but may provide a crucial time and energy advantage in the future.

However, these strategies are all still subject to debate in the scientific literature and definitely require further intensive, in-depth understanding of the formulation-dependent physicochemical processes within the cell before they can be applied on a scaled-up, industrial scale [48]. In this regard, the uncertain formation mechanism and composition of the SEI are major obstacles in understanding formation and aging. Studies illustrate that formation programs aiming at time optimization are in tension between minimizing impedance rise, improving capacity maintenance, and avoiding lithium plating [49]. Improper forming processes that are not matched to the conditions inside the cell can lead to uneven formation of SEI or undesirable lithium plating, which result in negative effects on lifetime due to premature capacity degradation and safety risks due to dendrite formation and short-circuit hazards [49], [50], [51], [52]. The development of advanced characterization techniques can contribute to further understanding of the formation, property formation, and aging behavior of SEI, and thus, represents the starting point of time-shortening or energy-saving measures.

As part of the manufacturing process, cell aging is an endof-line test that has the task of detecting irregular capacity degradation of individual cells over time, and thus, identifying defective cells [42]. Cost-reducing innovations are primarily concerned here with measurement technology and data-based modeling, which are able to detect defective cells more quickly, and thus, contribute to streamlining the overall process [53], [54], [55].

In addition to the possibility of innovating the existing process through technological innovations, a final way to be mentioned is to increase the performance of the cells through product innovations (in this case, optimization of the cell chemistry). This reduces the number of cells required, which indirectly contributes to an increase in efficiency. Solid-state batteries, although highly uncertain due to the lack of real production data, have already been addressed by bottom-up cost calculations and predict, competitive costs in the long term, at least for their subcategory of solid-state sulfide cells [56]. For oxide batteries, the energy-intensive processing step of sintering is considered an obstacle. The aerosol deposition method is shown to be promising and could bring cell technologies down to a cost level of up to \$150/kWh [57]. Implications for their processability and possible optimization potentials by increasing the throughput or reducing the residence time are also discussed [58]. Depending on the intended application, the cell designs can be varied. For example, particularly thick electrode layers, especially for highenergy applications in all-electric automobiles, could lead to a reduced BEV pack cost by an additional 8% [22]. In addition to the variation in the electrode properties, there is further potential for innovation in the design of current collectors. Recent examples consider scientific studies as well as announcements by automotive companies regarding multitab or "tabless" design" [59], [60].

The findings of investigations into the various subaspects of battery cell production are aggregated in studies on the modeling of production costs. For lithium-ion battery cell manufacturing, however, this is not a completely unaddressed topic; thus, earlier works can be referenced. One key dataset used by a variety of studies, particularly for building bottom-up models to replicate battery cell manufacturing costs, is the BatPaC database from Argonne National Lab. BatPaC is a freely available LIB design and cost model that allows the evaluation of different LIB designs and chemistries to be predicted using simulation. This is done based on laboratory data and pack-level metrics. By evaluating the cost of battery packs at specified production levels, it can be used to predict material and energy requirements and to identify opportunities for cost reduction [61], [62]. Duffner et al. [53] provided an excellent overview of the previous research landscape on battery cell manufacturing cost modeling, highlighting the BatPaC model as being the most influential database in this scientific field to date.

Previous studies have highlighted a variety of individual content facets on the part of the impact of cost sensitivity of the selected product or process innovations. However, we observed here that the sources and references cited in many studies contain outdated data or data that have not yet progressed beyond a bottom-up approach [21], [63]. Datasets based on real battery cell production planning data are urgently needed to validate these planning-conceptual approaches. In this study, we follow up on the recommendations for future research by Duffner et al. [64]. Thus, the use of quantitative models for cost calculation, the provision of model architecture and input data, as well as the thematization of novel technologies and calculation of a standard format (21700) are carried out. In addition, our work responds

to the call for additional integration of process technology innovations and validation through concrete planning/operational data of a cell manufacturing research factory [7].

Furthermore, since the field of battery cell manufacturing has high innovation dynamics, studies based on current datasets are urgently needed to advance the field and conduct research toward successful technology transfer. Studies indicate that it is more difficult for battery research, exemplified as a subdiscipline of energy technologies encompassing diverse knowledge sectors [65], [66], to integrate innovative technologies into production processes than in comparable manufacturing sectors. This is due to the additional coordination effort of knowledge and its transfer to other use environments (use environments, [67], [68]) (e.g., from research to industrial application, or even use in different end products) [69]. This underpins the relevance of the topic of battery cell manufacturing to the knowledge field of technology and innovation management. Furthermore, it provides a case [70] that is suitable for deriving generalizable conclusions for similar industries involving multisectoral knowledge fields [71]. Discussions on the manifestations of technological change in industry, such as battery cell manufacturing as a multisectoral industrial field have scarcely been discussed so far. The analysis of concrete case studies can offer an added value to built-up knowledge, such as the identification of correlations between innovations of different origins and their influence on production efficiency at the system level. Such insights are helpful in taking a holistic view without disregarding individual technological innovations and their specific contributions. For example, different inputs can often be substituted for each other. Technological innovations that influence production efficiency should be discussed here, as well as novel application conditions that influence the appearance of the end-product, i.e., the battery cell.

III. SETUP OF BATTERY CELL PRODUCTION (CASE STUDY)

In Germany, the federal government is completely funding the construction of a large-scale factory for lithium-ion battery cell production with more than 700 million €, solely for research reasons. Fig. 1 shows a rendered picture of the factory during the earlier planning phase. In this factory, namely the "Research Fab Battery Cells FFB," battery cells shall be produced on an industrial scale, production problems will be identified, and new production technologies will be tested [72]. By setting up this unique technology accelerator, Germany and Europe aim to catch up with their Asian counterparts, which dominate the global battery cell market today. The factory, which is run by the "Fraunhofer-Gesellschaft," will be technically capable of producing up to 7.0 GWh/a of electrodes. This is close to industrial scale (e.g., SK Innovation, Georgia-US, 11.7 GWh/a [73]; Northvolt Zwei, Salzgitter-Germany, 16 GWh/a [74]; and CATL, Erfurt-Germany, 14 GWh/a [75]). Therefore, the "Research Fab Battery Cells FFB" will probably be the largest demonstration factory of its kind worldwide. The factory, which is the basis for this study, is going to be located in Münster, in the west of Germany. Two large state-of-the-art coating lines



Fig. 1. Rendering of the German research fab battery cell (picture: Artvisu, artur krause).

will be located at the elongated side of the canal. Right next to it, the assembly and formation as well as aging and testing for different cell formats (cylindrical, pouch, and prismatic) will be located. As the mission of this research fab is to gain knowledge about the processing of battery cells, good comparability through the use of a standardized format (such as the 21700 cell) is beneficial toward achieving this goal and the same is also recommended by corresponding literature for the development of further insights into cost modeling and the identification of complex cause–effect relationships [6], [7]. The output of the assembly and forming lines, however, are much smaller than that of the electrode lines, capable of producing 30 cylindrical cells per minute.

The "Research Fab Battery Cell" is the perfect environment for a comprehensive, state-of-the-art production tear down and cost modeling for modern battery cell production. The authors of this study are part of the factory management team and have unique insights in all aspects of cost planning. These insights shall be shared here as far as possible to give researchers as well as managers a reliable theoretical base and primary data for further modeling.

For the production tear down (case study), the exact layout of the research fab was not used, but all data from different planning scenarios and planning stages were utilized to design a state-of-the-art battery cell factory. Battery production and, therefore, the layout common factories usually consist of the following three main phases:

- 1) electrode manufacturing;
- 2) cell assembly;
- 3) cell finishing.

The layout of a state-of-the-art factory (machines and dry rooms) for all three production phases is shown in Fig. 2. The factory has an output of 200 cylindrical cells per minute (ppm), which is approximately 1 GWh/a, depending on the battery cell chemistry. This is a common maximum number for production lines when producing cylindrical cells today. Most factories multiply the number of such "200 ppm production lines" to increase the overall output of the factory. For example, a 5-GWh/a factory will have five times the machines, as shown in Fig. 2. This multiplication has only minor effects on production costs though, as scale effects are minor.



Fig. 2. Layout of a state-of-the-art battery cell factory with an output of 200 cells per minute.

The factory space for storage is not taken into account, as this is a highly case-specific factor and does not result from technical reasons. Rather, it is a decision between risk-taking and economic capital lockup. In addition, the shown layout is not optimized for space saving or material flow. Rather, it shows the required space dimensions for each manufacturing step and

IEEE TRANSACTIONS ON ENGINEERING MANAGEMENT

TABLE I TECHNICAL BATTERY CELL PARAMETER

Parameter	Value	Unit
Format type	Cylindrical	_
Format size	21700	mm
Cell configuration	High power	_
Nominal voltage	3.6	V
Nominal capacity	2.33	Ah
Chemistry cathode	NMC622	_
Chemistry anode	Graphite	_
Coating thickness cathode	42	μm
Coating thickness anode	45	μm
Number of coated sides	2	_
Areal capacity cathode	2.0	mAh/cm ²
Areal capacity anode	2.2	mAh/cm ²
Separator	Polyolefine	_

makes the production and cost modeling in the following pages easier to understand.

In Table I, the technical specifications of a typical high-power automotive battery cell (used for the modeling) are shown. These are standard values of an average NMC622 battery cell with a cylindrical 21700 format, with the aim of achieving high power rather than high energy. Using a production output of 200 cells per minute, the hourly produced battery capacity is $P_{\text{produced}} = 100.80 \text{ kWh/h}.$

In Table II, the technical specifications of the production machines and equipment are shown for each production step as well as for the majority of the required infrastructure. In Table III, the relevant costs are shown. This is the major bottom-up dataset used for the cost modeling and scenario analysis later in the text. The data sources include the target price offers from production machine vendors and manufacturers, which were requested during the factory planning process. To ensure high accuracy of the economic and technical data, multiple target price offers were obtained and analyzed. The table shows the average values of two or more offers. In a few cases, the numbers had to be interpolated or extrapolated according to the right production machine size. These numbers are marked individually. As shown in Table II, production machines have different maximum production capacities. Fig. 3 shows the maximum producible output of battery cells and battery components in kWh/h for each production step according to the defined battery cell specifications.

At a first gaze, the production output is not homogenous. However, for economic production, the output of all processes should be at the same level. This is very difficult to realize, as various machines from different vendors are required and have only certain predefined standard machine sizes. Therefore, certain variations in the producible output along the production line are common. In the demo factory, the bottleneck is the assembly line, which has a maximum output capacity of 200 cells per minute (ppm). This is equal to a production output of approximately 100 kWh/h.

The subsequent bottleneck includes the coating and drying processes in electrode production. As coating, drying, calendaring, and slitting are continuous roll-to-toll processes, the production output of calendaring and slitting is limited, although the machines are capable of (much) higher productivity. To ensure



Fig. 3. Maximum producible output of battery cell components P_{\max} for each production step.

uninterrupted production, the outputs of mixing and vacuum drying are designed to be slightly larger by purpose, as these are processes with single large batches. Later in the article, we will discuss the effect of a perfect homogeneous production flow on production costs.

IV. METHODOLOGY AND COST MODEL

Hueber et al. [77] divided the existing research landscape on battery cell manufacturing cost modeling into qualitative (intuitive and analogy based) and quantitative methods (parameter based and analytical). Analytical models are generally considered to have a higher level of detail and transparency [7]. Certain case-based approaches to the further development of cost models focus on their generalizable applicability across industries. For example, the case-based reasoning method is advantageous in terms of its ability to accept unknown information, take into account the results, and the ability to process certain cases [78]. Other cost-analytic studies deliberately focus on the uniqueness of construction and design practices of different fields and encourage recognition that the economic competitiveness of any technological choice depends on the context of its application [79]. Related to the context of battery cell manufacturing, analytical bottom-up models have proven to be useful in illustrating the cost-causing structures of processing. A widely used approach is the process-based cost modeling (PBCM) technique [7], [13], [19].

Studies have addressed the general shortcoming that while the most effective way is to derive cost positions from the design of the products and processes themselves, cost data are rarely available for product/process designers in a usable form [80]. Thus, the analysis of the study presented here is fundamentally based on a bottom-up approach to cost modeling, which involves accounting for the influence of a wide variety of processing and product design variables within the calculation. Previous works have suggested a PBCM-based approach for calculation

DEGEN AND KRÄTZIG: DEGEN AND KRÄTZIG: MODELING LARGE-SCALE MANUFACTURING OF LITHIUM-ION BATTERY CELLS

Required worker Dry room Appendix: max. performance Process Process Machine/Process Investment costs Electric energy Gas energy Dimension Areal Max. producible number footprin step (-) [-] consumption (E_{electric}) [kWh] output (Egas) [kWh] (Pmax) [kWh/h] (IC) [€] (w×l) [m×m] (j) [-] (-) [-] (A_{ft} nt) [m²] (n_{worker}) [-] parameters 72.0 302.38 250 l/h Electrode Batch mixer 1,080,000 20 8×9 no 1a manufacturing 2a Coater and drver 5,500,000 75 825 7×60 420.0 179.63 no $30 \text{ m/min} \times 0.7 \text{ m}$ 2 (anode) 3a Calender 2,900,000 60 4.5×16 72.0 228.10 no 30 m/min × 0.8 m 4a Slitter 1,150,000 45 4.5×16 72.0 570.24 80 m/min × 0.75 m no 267.78 5a Vacuum dryer 1,200,000 5 210 6×8 48.0 3x4 Coils, a 4900 m partly Electrode 1b Batch mixe 1.235.556 20 296.19 250 Uh no manufacturing (cathode) 2h Coater and dryer 5,500,000 75 825 7×60 420.0 163.30 30 m/min × 0.7 m no 3b 2.900.000 60 4.5×16 72.0 207.36 30 m/min × 0.8 m Calender no 1,150,000 45 4.5×16 72.0 518.40 80 m/min × 0.75 m 4b Slitter no partly 5h Vacuum drver 1.200.000 5 210 6×8 48.0243.43 3x4 Coils. a 4900 m Assembly 6 Winding machine 100.80 200 ppn yes 160 5³⁾ 100.80 200 ppm Assembly machine 9,000,0003) 8×80³⁾ 640.0 3) yes 8 Washing machine 200 100.80 partly 200 ppm 0 25 Formation Soaking rack 100.80 4h, 200 ppn no 10 1000 20×24 480.0 100.80 20h, 200 ppm Formation racks no 11 40 Aging High Temp. aging racks 1.5×13.3 20.0 100.80 no 24h, 200 ppm 45,000,0002). 4 12 Room Temp. aging racks 24×30 720.0 100.80 no 504h, 200 ppm 100 2×7.5 100.80 1h, 200 ppm Testing 13 OCV test machines 15.0 no 200 ppm 14 Material handling machines 200 100.80 Miscellar no Dry room Dry room 5,000,000 163 913 12×80 960 n/a -40°C dew point Building /shop floor 5200 Building 11,440,000 n/a n/a

 TABLE II

 MACHINE AND PRODUCTION PROCESS RELATED COSTS (REFERRING TO [76])

1) Values interpolated, 2) Values extrapolated, 3) Data for entire set.

TABLE III FURTHER COSTS AND INFORMATION

Parameter	Variable	Value	Unit	Note
Labor costs	PC _{worker}	40.00	€/h	For location Germany
Energy costs (electricity)	CC _{electric}	0.151	€/kWh	For location Germany
Energy costs (gas)	CC_{gas}	0.023	€/kWh	For location Germany
Area unit costs	$AC_{building}$	0.008	€/m²/h	For location Germany
Overhead rate	ioverhead	60	%	Based on personnel costs
Production time	-	8760	h/year	Quasistatic $24h \times 7d$ production
Deprecation time machines	$t_{depreciation_m}$	10	years	
Deprecation time building & dry rooms	$t_{depreciation_b}$	30	years	

of battery cost [7], which we validate via the analysis of a consistent data set from the real planning work of a battery cell production and complement as well as with a discussion on the cost impact of emerging technologies [80]. In this PBCM technique, a bottom-up approach introduced by [79], costs are modeled in the following three steps:

1) process model;

- 2) operations model;
- 3) financial model.

In the process model, the product and its features that should be manufactured are transferred into a technical production layout and its technical parameters (Fig. 2 and Table II). In the operations model, the operating conditions are transferred into resource requirements (Tables II and III). In the financial model, the resource requirements are then transformed into financial KPIs. This is done as follows: To calculate the battery cell production costs in \notin /kWh, we propose the following calculation:

$$C_{\text{total}} = \sum_{j=1}^{n} C_{\text{direct}}^{j} + C_{\text{indirect}}.$$
 (1)

First, we differ between direct and indirect costs. All costs are allocated to one produced kWh battery capacity (\notin /kWh). The direct costs are allocated to each individual manufacturing process (*j*) for a continuous output of 200 cells per minute (respectively 100.80 kWh/h). The direct costs are calculated as follows:

$$C_{\text{direct}}^{j} = C_{\text{machine}}^{j} + C_{\text{energy}}^{j} + C_{\text{labor}}^{j} + C_{\text{footprint}}^{j}.$$
 (2)

In direct costs, material costs are not taken into account, as these costs are highly individual and can significantly affect the cost model. Material costs can be added later by each individual researcher, if required. The indirect costs are calculated as follows:

$$C_{\text{indirect}} = C_{\text{building}} + C_{\text{dry room}} + C_{\text{overhead}}.$$
 (3)

The equations for the detailed calculation can be found in the appendix. As shown in Fig. 2, the entire factory is planned in the spirit of lean production. Thus, costs that do not result in value creation are minimized.

In this article, all relevant input data and values are named in absolute numbers. Thus, each researcher can recalculate the presented results using them. More importantly, each researcher

IEEE TRANSACTIONS ON ENGINEERING MANAGEMENT



Fig. 4. Structure of total costs (C_{total}) for battery cell manufacturing (basic case).



Fig. 5. Structure of direct costs (C_{direct}) for battery cell manufacturing for each production process (basic case).

can use the values of this basic factory layout to modify, extend, or update their own models.

To demonstrate the capability of this detailed model and of the given dataset, the following will be calculated, presented, and analyzed:

- 1) costs with reference production (case study);
- 2) costs with optimized production;
- 3) validation/comparison of calculated costs;
- 4) costs with future technologies.

First, we will analyze how much the manufacturing of the battery cell will cost and what are the major cost drivers with the presented factory. Second, the impact of optimizing production and production locations will be analyzed. In the third step, these results will be compared and validated with other and known cost calculations. Fourth, the impact of novel cell and production technologies on industrial readiness will be analyzed for the forthcoming years. Through this, we demonstrate not only the validity of the model and the dataset but also the capability to predict the state-of-the-art costs as well as future battery cell costs.

V. RESULTS AND DISCUSSION

A. Costs With Reference Production (Case Study)

For the basic case, we assume that the factory is located in Germany, Europe. As Germany is located in the heart of the European market and is also a major player in the automotive industry, many cell factories are being built there at the moment (production capacity of up to 500 GWh/a has already been announced for 2030 [3]). However, Germany has high labor costs, high energy costs, and high building costs. Fig. 4 shows the cost structure of C_{total} . It is shown that in this production plant, the production of 1 kWh battery capacity costs 26.92 \in . The main cost drivers are costs for machine depreciation, followed by labor and energy costs. The costs for the footprint and building are negligibly small. Thus, even when taking further storage space into account, which was not done here, this would most likely have no significant effect on the total cost.

Fig. 5 shows the cost structure of C_{direct} for the different production steps. Costs for multiple machines have already been considered in the cost structure. It is shown that cell formation is the production step that makes the highest contribution to cost, followed by coating and drying, and then assembly. The reason for this cost peak is that the formation and assembly consist of many single process steps, such as electrode manufacturing, which are summarized here. Nonetheless, formation contributes approximately 45% to the overall direct cost, which is quite large. However, here we are analyzing only the basic case, and further improvement will be taken into account in the following sections.

B. Costs With Optimized Production

As already indicated, there are certain general possibilities for improving the production of battery cells. These include the following:

1) homogenize the production output;

DEGEN AND KRÄTZIG: DEGEN AND KRÄTZIG: MODELING LARGE-SCALE MANUFACTURING OF LITHIUM-ION BATTERY CELLS



Fig. 6. Effect of optimization measures on battery cell production costs.

- 2) reduce investment costs for machinery;
- 3) location with lower energy costs;
- 4) location with lower labor costs.
 - 1) As shown in Fig. 3, the output of the production processes along the manufacturing chain is not perfectly homogenized. This means that a full capacity load is not achieved for each individual process step. It would be desirable to optimally coordinate the process steps with each other to achieve maximum individual capacity utilization and possibly avoid excessively high investment costs. This is theoretically possible, though, when buying machines in high numbers; for example, to equip a factory with >10 GWh/a production output. To calculate this effect, the output was leveled at exactly 200 ppm here (= 100.80 kWh/h). For calculation purposes, the machine investment costs and energy consumption were linearly interpolated. As we are discussing only about minor changes in machine size, a quasilinear approach is allowed from our perspective. The staff number and footprint are kept the same.
 - 2) Besides that, there is still potential, to decrease the investment costs for machinery further. The battery cell production described in the case study in this article is currently still under development. The investment costs listed in Table II are the prices obtained at the beginning of the negotiation phase. As negotiations progressed, different price reductions in the range of 5%–20% have been achieved so far. Differences lie in the offer of standardized products, such as analytical equipment, for which higher price reductions could be negotiated than for individual products. One reason may be that the supplier's personnel costs are higher for such products, as there is a greater need for adaptation to customer requirements. According to our sources, entire production lines (electrode manufacturing, assembly, forming) with a capacity of 200 ppm can be purchased for 60–80 million €. Regarding total average costs, we have, therefore, made the experience that a 10% discount compared to the initial price on the investment costs for machines is feasible.

TABLE IV Measures to Optimize Cell Production

No.	Measure	Basic	Optimized
M1	Machine outputs	Variable	Pproduced
M2	Machine costs	100%	90%
M3	Electricity costs	0.151 €/kWh	0.069 €/kWh
M4	Labor costs	40.00 €/h	8.50 €/h

- 3) Energy is a major cost driver in battery cell production. Thus, many cell factories are built in countries with low energy costs. Examples for Europe are the Scandinavian countries, particularly Norway, with electricity costs of 0.041 €/kWh [81]. However, labor costs are also quite high, which is why an alternative location might be China, where electricity costs 0.069 €/kWh [82].
- 4) In addition, labor costs constitute a major part of the overall cell cost, especially when analyzing a factory at a location in a high-wage country in the heart of Europe. Thus, by relocating the factory to a country with a lower labor cost, cost reduction is possible (labor costs in China: 8.50€/h) [83]. Table IV shows the quantitative assumptions and measures for an optimized battery cell production scenario located in China instead of Germany. Fig. 6 shows the effects of cost reduction measures on the battery cell cost.

It is shown that by optimizing M1 (production output homogenization) and measure M2 (machine purchase optimization), production costs can be reduced, but only to a minor extent. The most significant effect can be realized by reducing energy costs (M3). By optimizing M1, M2, and M3, the cost of cell production can be reduced significantly to $22.75 \notin kWh$. Finally, by reducing the labor costs (M4), further reductions can be realized. By applying these measures, a cell production cost of $12.25 \notin kWh$ can be achieved today.

C. Validation

All aforementioned numbers were calculated using the presented bottom-up model and the given input data. To validate the model and the numbers, the calculated results are compared

 TABLE V

 Reported Battery Cell Cost (INCL. Material)

NMC battery cell cost	Year	Source
93 €/kWh; 112 \$/kWh	2019	[84]
88 €/kWh; 106 \$/kWh	2020	[7]
85 €/kWh ; 102 \$/kWh	2020	[8]
83 €/kWh ; 99 \$/kWh	2020	[85]
77 €/kWh ; 92 \$/kWh	2020	[3]

TABLE VI Reported Cell Production/Material Cost Ratio

Production/Material	Year	Source
26% /74%	2016	[86]
25% /75%	2018	[6]
30% /70%	2018	[87]
30% /70%	2019	[88]
35% /65%	2019	[89]
35% /65%	2020	[3]
20% /80%	2020	[9]
23%/77%	2019	[84]
23%/77%	2021	[7]

with known price calculations from the research community and market research. Before doing so, it must be taken into account that the following cost drivers, besides the material, are not considered in our presented model:

- 1) scrap rate;
- 2) larger maintenance costs;
- 3) larger maintenance time.

The reason why these are not considered here is that the named factors differ strongly between manufacturers and are difficult to quantify reliably. Thus, realistic production costs are probably slightly higher than the costs calculated via the proposed model. We estimate that the calculated prices for battery cell production are about 10% higher when taking scrap rate and larger maintenance into account.

Many battery costs available publicly are for the cell or pack level, taking into account production costs and material costs [7], [21]. State-of-the-art costs for battery cells are slightly above 100 USD, although 2020 was the first year when costs below 100 USD were reported; see Table V.

Selected studies, in addition to summing up the total costs of cell manufacturing at the cell level, also address the breakdown between material, process, and other costs. There are values given in the literature that describe the ratio of battery cell cost between material and production costs. Common ratios are shown in Table VI.

Although the ratio differs from source to source, a proportion of manufacturing cost of 20%–35% seems to be common. By combining the average battery cell cost and the production/material cost ratio, a cost range of 14.40–32.55 €/kWh for battery cell production is identified. This is quite exactly the cost we calculated with our model (optimized to basic scenario: 12.25–26.92 €/kWh). However, in our optimized scenario, we identified slightly lower production costs. But we did not consider the scrap rate in our calculations. In general, there is a strong match between our bottom-up based results and those of the state-of-the-art literature and market research. Against this background, our cost model seems to be valid and covers the latest state-of-the-art production technology.

D. Costs With Future Technologies

After having shown that the model provides results that fit to common cost values, we shall investigate how battery cell production and cost structure can look like in the near future. For this, not just theoretical ideas (e.g., increasing coating thickness), but genuine aspirations of the manufacturing industry are analyzed. The most important approaches today include the following:

- 1) NMC811 & SiO/C cell chemistry;
- 2) tabless electrodes and continuous coating;
- 3) dry coating technology.

NMC811 cell chemistry for cathodes is currently replacing the NMC622 cathode chemistry. This is an ongoing process, and the first battery cells with NMC811 cathodes are already on the market. Thus, NMC811 represents current state-of-the-art to the best of our knowledge [3]. In terms of production costs, a major benefit is the higher specific capacity of NMC811 (200 mAh/g) than that of NMC622 (170 mAh/g). In addition, the material is cheaper because of the reduced amount of cost-intensive cobalt required. The challenges in using NMC811 in production are related to the electro-chemistry and high requirements for the dryness of the production environment. However, recent studies highlight the massive potential of "cobalt-free" lithium-ion batteries for a more sustainable manufacture of electric vehicles in the coming decades [90].

In addition, the cell chemistry on the anode side has been further developed to SiO/C cell chemistry with a silicon oxide content of 5%. The composite electrodes lead to increased volumetric energy densities, since SiO has a specific capacity (2400 mAh/g) that is almost seven times higher than that of C [91], [92], [93]. By utilizing it, the specific capacity of the anode (graphite: 360 mAh/g) can be increased (SiO/C with 5% SiO: 460 mAh/g). The SiO/C layer thickness of the anodes can be designed thinner and/or the cathodes thicker depending on the application (HE/HP). In our future scenario F1, we assume the same physical design of the cell as shown in Table I, but with NMC811 on the cathode side and SiO/C (with 5% SiO) on the anode side. With regard to the production costs, it can be stated that less material is required per kWh produced. For such a cell, with a production output of 200 cells per minute, $P_{\text{produced}} =$ 118.6 kWh/h of battery capacity can be produced instead of $P_{\text{produced}} = 100.8 \text{ kWh/h.}$

Another approach to reduce production costs is multitab or "tabless" battery cell production and cell design. A tabless design involves cylindrical battery cells, which provide benefits in terms of battery cell performance and production performance. Studies report of benefits in cell performance in multitab instead of single-tab design by +23% SoC fast charging efficiency [59]. Using tabs in cylindrical cells limits their production in two ways. The first is the limitation of the tab-welding process in cell assembly. Second, and even more important, is the limitation of coating during cell production. For a cylindrical cell design with tabs, it is required that the electrodes have free surface

areas that are not coated. This coating has to be interrupted frequently, which limits the coating velocity considerably. The coater, shown in Table II, has a maximum coating velocity of $v_c = 30$ m/min in discontinuous mode and $v_c = 80$ m/min in continuous mode. Thus, using a tabless design, the coating velocity can be almost tripled. However, a new bottleneck related to the length of the dryer (here, l = 40 m) occurs. When the coating velocity is increased and the drying time is kept constant, the dryer length must be increased. For a 40-m long dryer, coating velocities of $v_c = 50$ m/min are possible. However, for a coating velocity of $v_c = 80$ m/min, an extended dryer length of about 70 m is required.

In our future scenario F2, we increase the coating velocity from $v_c = 30$ m/min to $v_c = 50$ m/min. This affects the required machine size, investment costs, and energy consumption. Similar to M1, staff number and footprint are kept the same. Moreover, the data of the assembly line are kept the same. This is because tab welding is not a major bottleneck in assembly. Although tab welding is eliminated, a connection still needs to be generated by another machine. Therefore, the effect of the tabless design on the assembly costs is probably negligible. Since tabless design is an innovation in cell design, the corresponding advantages are to be found at the cell level. While material innovations primarily aim at the realization of higher energy densities, cells with tabless design enable the realization of higher power densities. This can be explained by the fact that there are shorter diffusion paths of the electrons to the current collectors of a cell, and thus less heat development and other attractive electrical effects occur inside the cell. This is indicated by previous studies on tab designs [94]. In addition to the cell-level benefits, the application of tabless design also enables the realization of increased volumetric energy density at the pack level by reducing the free space between cylindrical cells. A recent, prominent example that is announced to use tabless design is the codeveloped 4680 cell from Panasonic and Tesla [2]. However, the fundamental cell design is not new, since examples of commercially available cells from the past can be found, e.g., SAFT's VL6P-cell, patented in 2005 [95].

Another future battery cell technology is dry coating [96]. As shown in Fig. 5, coating and particularly drying constitute the main cost driver in electrode production. By dry coating, no extensive NMP solvent recovery is required and neither energy-intensive nor space-intensive drying is required. However, dry coating technology is still in development. One of the technology leaders is probably Maxwell Technologies, Inc., which actually originates and gained experience from the production of capacitors. Other dry coating technologies are, e.g., the dry transfer process [97], the dry extrusion process [98], the fluidized bed process, isothermal pressing [99], various deposition techniques [100], etc.

In our future scenario F3, we assume that in dry coating, the investment costs for coater, dryer, and NMP recovery are halved. Dryer and NMP recovery are no longer needed but machine manufacturers will also use technological leadership for high prices. We believe that dry coating machines require slightly more electricity ($E_{\text{electric}} = 50 \text{ kW}$) due to the higher forces

during coating; however, no gas is required for drying ($E_{\text{gas}} = 0 \text{ kW}$). In addition, the areal footprint is significantly reduced from $A_{\text{footprint}} = 420 \text{ m}^2$ to $A_{\text{footprint}} = 70 \text{ m}^2$.

In Fig. 7, the effects of the future scenarios F1 (new cell chemistry), F2 (tabless design), and F3 (dry coating) are shown. The basis for the calculations is the optimized scenario with applied improvement measures M1, M2, M3, and M4.

It is shown that by using an NMC811 & SiO/C cell chemistry, the production costs per kWh can be reduced significantly by almost 17%. Even though the absolute production costs are the same, the relative costs per kWh can be decreased owing to the higher energy density of the cell. With such a chemistry, a cell production cost of $10.15 \in /kWh$ is feasible.

By using a tabless cell design and NMC622 & C chemistry, the production costs can be reduced by approximately 5% to $11.55 \notin k$ Wh. This is because a tabless design mainly reduces the coating cost in production. This has a positive effect, but there are also other cost drivers in the overall production, which limit the overall effect.

By using dry coating, further cost reduction is possible. In comparison to standard wet coating with NMC622 & C chemistry, production costs can be decreased by 6%–7% to $11.34 \notin$ /kWh. This is for the same reasons as those when using the tabless cell design for production.

When combining all three future measures (F1, F2, F3), according to our calculations, production costs can be decreased to as much as $9.52 \notin k$ Wh, a 21% decrease over current prices predicted by our model. Of note, it is remarkable that the effect of tabless design and dry coating is only minor when using NMC811 chemistry ($\Delta C_{total} = 0.63 \notin k$ Wh). Nonetheless, it has to be taken into account that both measures not only have an effect on the production costs but also on the battery cell performance/parameters.

With a tabless design, e.g., the thermal load during charging and discharging is lower. This positively affects C-rates, lifetime, etc. In addition, larger cell diameters are possible, which again positively affects production costs. Similarly, dry coating technology significantly reduces the energy demand in production, and therefore, positively affects the CO₂ footprint. This has become increasingly important from a societal and political perspective. However, having said this, the major outcome here is that, by our calculations, cell production costs at around 11.00 ℓ/kWh appear to be realistic within the next few years. Fig. 8 shows the breakup of these costs, and the direct cost C_{direct} in detail.

Here, only the direct $\cos (C_{\text{direct}} = 8.12 \notin kWh)$ is shown. The indirect $\cos (C_{\text{indirect}} = 1.41 \notin kWh)$ is less relevant. It is shown that formation is by far the largest cost driver. In particular, the machine depreciation $\cos (C_{\text{machine}})$ and energy $\cos (C_{\text{energy}})$ are here to name. Approximately 1000 formation racks are required to form an output of 200 ppm for a duration of 20 h. While these are high investment and energy costs on the one hand, on the other hand, the lead time is very high (compare Table II). Thus, the cost per kWh is also high, as no real-scale effects can be obtained. Thus, intensive focus needs to be devoted to formation for further reducing battery cell production costs in future.

IEEE TRANSACTIONS ON ENGINEERING MANAGEMENT



Fig. 7. Effect of future production technologies on battery cell production costs.



Fig. 8. Structure of battery direct cost (C_{direct}) for battery cell manufacturing (future cases: F1, F2, and F3).

E. Critical Discussion

We have shown that our cost model and our dataset are suitable for reliably calculating the state-of-the-art battery cell production costs. We have also shown that our results fit very well with the latest cell production costs sourced from literature and market research. Nonetheless, there are certain issues that are not taken into account within the model, which should be mentioned and assessed here.

As already mentioned, the scrap rate and larger maintenance costs are not considered in our model. Both vary highly from case to case and depend on the production skills and know-how. Besides, there are no reliable data available for the same. The common rates we notice in discussions are 5%–10% but without any evidence. Thus, we decided to not take the scrap rate into account. In addition, we did not consider larger maintenance costs, as this also varies from case to case and standard maintenance is already considered in the overhead costs.

In our lowest-cost scenario, the factory is located in China. Due to attractive cost structures, the scenario of processing battery cells in the Far East and subsequent international logistics has been pursued on a large scale in the past. The transformation of the automotive industry toward electric mobility underlines the associated need to keep the added value of the final product within the corporation. Announcements of the manufacturers for the conclusion of a significantly higher demand for battery cells to be manufactured locally, close to the customers. Compliance with the legally stricter regulations and standards of the automotive industry in the direction of sustainable manufacturing also suggests this development.

In such a case, new production technologies, such as dry coating, will have far higher economic benefits than those shown here. This also needs to be taken into account when working with our cost model and assessing results. Because the energy demand for dry coatings is generally very less compared to conventional wet coatings, the technology allows production activities to be more independent of the respective local energy costs. As a result, the technology has the potential to establish itself globally, which could lead to its further development, leading to even lower per-unit costs.

With increasing focus on the environment by the society, the topic of "green" manufacturing has become increasingly important. In Germany, CO₂ emission certificates will cost $55 \notin /t$ -CO₂-eq in 2025 [101]. As battery cell production requires a high amount of energy, not just electricity, but also natural gas, this will be a significant future cost driver. This also has to be considered when working with the proposed model. We did not do so because emission regulations vary widely across the world.

Finally, yet importantly, we want to underline that the future production technology approaches we have presented have primary benefits in terms of cell performance parameters. While they affect production costs positively as well, they do so only in a minor way. Thus, this study should not cause doubts or hesitations in terms of new production technologies. Especially in high-wage countries and countries with expensive but "green" energy, these new technologies hold promise to be "game changers" in terms of production costs.

F. Future Outlook

Based on our analysis, we see two main levers to decrease battery production costs: 1) improving the cell finishing process and 2) improving cell chemistry. Both are research topics related to chemistry rather than engineering. While the interaction of the individual components within the cells is optimized in research work on more efficient cell chemistries in order to realize ever higher energy and power densities as well as lifetime, a distinction must be made in cell finishing between the formation process and cell aging. In the formation process in particular, the key knowledge lies in the sequence of a recipe-specific formation program that is aligned to the interaction between applied current, time, and temperature with the chemical processes inside the cell, defined by the material use of additives, electrolytes and the cell format per se, to form a stable boundary layer, the SEI [39], [47].

The formation of a stable SEI is largely determining the quality of the final cell not only in terms of energy and power density but also regarding lifetime and safety issues [6], [40]. While the optimization work on the formation program has rather low requirements for innovative hardware, the potential for energy savings through the introduction of advanced power electronics has been reported though. Former studies report a high energy consumption in the formation process that is not only due to the thermodynamic formation of the SEI, but also to dissipative energy losses during charge/discharge cycles and cooling to maintain a constant temperature and homogeneity between all cells [76].

In cell aging, the cost-causing (Cpex for aging infrastructure, footprint of aging infrastructure) problem arises from the enormous amount of time (3 weeks) that cells are tested for irregular capacity degradation. Here, innovative measurement methods, e.g., advanced self-discharge measurement technology [102], can help to detect defective cells at an early stage, to streamline the overall manufacturing process by selling the cells faster and thus to significantly reduce the need for infrastructure (extensive storage facilities).

Furthermore, automation and the setup of digitization techniques are important measures that can be taken to reduce failure probability, understand causal relationships between processes and can reduce personnel costs further, especially in high-wage countries [103]. Another measure is to reduce the amount of energy required for production. CO₂ emissions will become more expensive and will also be legally restricted in the future. These are future production costs that need to be considered today. In addition to this technology-oriented outlook, we would like to provide an outlook of what can be done with our basic model in the future. Suggestions for further research are as follows:

- 1) life cycle assessment of battery cell production;
- 2) economic assessment of different production locations;
- assessing the economic impact of new production technologies and new cell technologies;
- 4) battery cell cost scenario analyses;
- enhancement of the model by adding new or further improving parameters;
- 6) identification of future cost drivers.

We hope that our model and, particularly, the dataset contribute to the research community involved in fields related to product manufacturing, especially battery-related research and production management.

VI. CONCLUSION

In this article, we presented unique insights into battery cell production, including a corresponding technical and economic dataset as well as a cost calculation model capable of illustrating cost-cutting production scenarios for evaluating contributions of new and emerging technologies to business strategy alike [104]. Based on real battery cell factory plans, we derived a state-ofthe-art demo factory setup, both technically and economically. For this battery cell factory, we showed the calculation of the battery cell production cost per kWh for different scenarios. We showed that our calculated battery cell production costs match very well with the reported production costs sourced from the latest market research. Thus, we were able to validate the model as well as the factory setup and dataset. Subsequently, we analyzed how battery cell production costs can be decreased further in the near future by using new technologies and cell chemistry. We identified a minimum cell production cost level of approximately 14.00 €/kWh with currently established technology and approximately 11.00 €/kWh with latest/upcoming technology.

Our work contributes to reducing complexity related to the integration of new and emerging technologies by methodically disclosing the cost structure of battery cell production at system level in the sense of an anticipatory technology assessment. We add to the methodology of technology assessment by extending PBCM technique with cost-reducing production scenarios and testing their applicability using real planning data. From our analysis, we can derive implications for strategic technology planning by linking the technology planning mode with the respective level of technological maturity. We can confirm former research results, which relate an increased planning risk for architectural and radical design changes to the product and process system level of battery cell production (e.g., dry coating), since an increasing complexity (also on product level) results out of multiple fields of knowledge involved [105]. We contribute to the knowledge field of technology management by showing which key performance indicators are applied at different dimensions (cell, process, and system level) within battery cell manufacturing and how these are interrelated. By applying the cost model, we showed the impact of current technological innovations in

battery research on process design. In doing so, we bridge the gap between research efforts to optimize product properties at the battery cell level to production-related quality criteria and contribute a transferable case study for cross-disciplinary innovations.

Apart from further broadening the subject of cost structures in battery cell manufacturing, another cornerstone of the investigation at hand relates to the disclosed factory layout/setup and corresponding primary dataset. Based on this tangible dataset of real battery cell production, practical implications can be derived. Using these results, depending on their discipline, researchers would be able to, for example, better allocate the need for investigations on specific phenomena from the field of engineering and technology management to the subject of battery cell production, or direct their activities in the field of materials research in a more targeted, that is, exploitation-related manner.

Implications for managers engaged in the battery cell manufacturing industry are that our approach to cost modeling and the datasets of battery cell manufacturing addressed here serve as a basis for benchmarking own activities and measures. As such, a more profound knowledge of the reference production configuration can contribute to successfully increasing production efficiency as well as integrating novel products and services. The findings of our study will enable policymakers to draw conclusions about the cross-relationship between innovations at the material, process, and system levels and the development of funding programs that are tailored to meet these needs for future technology development more effectively.

In a broader scope, our case reflects a multifaceted example on how the role of technological change in industries involves multisectoral knowledge fields and emphasizes the need to adapt institutional structures and processes for promoting innovation more flexible, yet optimization driven, and guiding toward ambidextrous innovation strategies. The study is based on a dataset of a real case study representing battery cell production during development. Thus, the scope of this study is limited to the field of battery cell production. The linking of aspects of product design with those of process design may be completely different in other manufacturing industries. Subsequent studies should attempt to draw on the theoretical base of this article to abstract more generalizable methodological approaches. Further follow-up studies may seek to incorporate learning and optimization efforts with respect to actual plant operations, evaluation of their effectiveness, and retrospective observation of the effects of process innovations on cell costs. Since the boundaries between economic and ecological/social realms are becoming increasingly blurred, especially with regard to the different dimensions of sustainability [106], a subsequent study that extends the approach presented here with corresponding KPIs would be desirable.

APPENDIX

Exchange rate for all calculations: $1.00 \in = 1.20 \text{ US}$ Assumption: production runs 365 days a year, 24 hours a day

$C_{machine}^{j} = \frac{1}{t_{dep}}$	$\frac{IC^{j}}{reciation_{m} \cdot P_{produced}}$
IC ^j	investment costs [€]
t _{depreciation_m}	machine depreciation time [h]
	here: 87 600 h (10 years)
$P_{produced}$	hourly produced battery power [kWh/h]
	here: 100.80 kWh/a (200 cells per minute)

$$C_{energy}^{j} = \frac{E_{electric}^{j} \cdot CC_{electric}}{P_{produced}} + \frac{E_{gas}^{j} \cdot CC_{gas}}{P_{produced}}$$

$$E_{electric}^{j} \qquad \text{energy consumption of electricity [kWh]}$$

$$E_{gas}^{j} \qquad \text{energy consumption of gas [kWh]}$$

$$CC_{electric} \qquad \text{consumption costs for electric energy [€/kWh]}$$

$$CC_{gas} \qquad \text{consumption costs for gas energy [€/kWh]}$$

$$C_{labor}^{j} = \frac{n_{worker}^{j} \cdot PC_{worker}}{P_{produced}}$$

$$n_{worker}^{j} \qquad \text{number of required workers [-]}$$

$$PC_{worker} \qquad \text{personnel costs [\mathcal{C}h]}$$

$$here: 40.00 \ \mathcal{C}/h \ (Basic)$$

$$C_{footprint}^{j} = \frac{A_{footprint}^{j} \cdot AC_{building}}{P_{produced}}$$

$$A_{footprint}^{j} \qquad \text{area of machine footprint [m2]}$$

$$AC_{building} \qquad \text{area unit cost [} \pounds/m^{2}/h]$$

$$here: 0.008 \notin/m^{2}/h$$

$$C_{Building} = \frac{\left(A_{building} - \sum_{j=1}^{n} A_{footprint}^{j}\right) \cdot AC_{building}}{P_{produced}}$$

$$A_{building} \quad \text{area of entire building [m2]}$$

here:
$$5200 \text{ m}^2$$

$$C_{dry\,room} = \frac{IC^{dry\,room}}{t_{depreciation} \cdot P_{produced}} + \frac{E_{electric}^{dry\,room} \cdot CC_{electric}}{P_{produced}} + \frac{E_{gas}^{dryroom} \cdot CC_{gas}}{P_{produced}}$$

$$C_{overhead} = \sum_{j=1}^{n} C_{labor}^{j} \cdot i_{overhead}$$

$$i_{overhead} \quad \text{overhead rate [-]}$$

$$here i=0.6$$

REFERENCES

- World Economic Forum, "A vision for a sustainable battery value chain in 2030: Unlocking the full potential to power sustainable development and climate change mitigation," Insight Report, World Economic Forum, Cologny/Geneva Switzerland, 2019, Accessed: Jan. 20, 2021. [Online]. Available: http://www3.weforum.org/docs/WEF_A_Vision_for_ a_Sustainable_Battery_Value_Chain_in_2030_Report.pdf
- [2] E. Musk, "Tesla battery day 2020," Fremont, CA, USA, Sep. 2020.

- [3] S. Michaelis and E. Rahimzei, "Roadmap batterieproduktionsmittel 2030: Update 2020," VDMA, Frankfurt am Main, Germany, 2020, Accessed: Jan. 20, 2021. [Online]. Available: https://battprod.vdma.org/en/ viewer/-/v2article/render/59579372
- [4] B. Nykvist and M. Nilsson, "Rapidly falling costs of battery packs for electric vehicles," *Nature Climate Change*, vol. 5, no. 4, pp. 329–332, 2015, doi: 10.1038/nclimate2564.
- [5] R. Schmuch, R. Wagner, G. Hörpel, T. Placke, and M. Winter, "Performance and cost of materials for lithium-based rechargeable automotive batteries," *Nature Energy*, vol. 3, no. 4, pp. 267–278, 2018, doi: 10.1038/s41560-018-0107-2.
- [6] A. Kwade, W. Haselrieder, R. Leithoff, A. Modlinger, F. Dietrich, and K. Droeder, "Current status and challenges for automotive battery production technologies," *Nature Energy*, vol. 3, no. 4, pp. 290–300, 2018, doi: 10.1038/s41560-018-0130-3.
- [7] F. Duffner, L. Mauler, M. Wentker, J. Leker, and M. Winter, "Large-scale automotive battery cell manufacturing: Analyzing strategic and operational effects on manufacturing costs," *Int. J. Prod. Econ.*, vol. 232, 2021, Art. no. 107982, doi: 10.1016/j.ijpe.2020.107982.
- [8] BloombergNEF, "Battery pack prices cited below \$100/kWh for the first time in 2020," While Market Average Sits at \$137/kWh, Accessed: Jan. 20, 2021. [Online]. Available: https://about.bnef.com/blog/batterypack-prices-cited-below-100-kwh-for-the-first-time-in-2020-whilemarket-average-sits-at-137-kwh/
- [9] P. Lamp, "BMW kompetenzzentrum batteriezelle," Batterieforum Deutschland, Berlin, Jan. 2020.
- [10] L. Mauler, F. Duffner, and J. Leker, "Economies of scale in battery cell manufacturing: The impact of material and process innovations," *Appl. Energy*, vol. 286, 2021, Art. no. 116499, doi: 10.1016/j.apenergy.2021.116499.
- [11] P. A. Nelson, S. Ahmed, K. G. Gallagher, and D. W. Dees, "Cost savings for manufacturing lithium batteries in a flexible plant," *J. Power Sources*, vol. 283, no. 1, pp. 506–516, 2015, doi: 10.1016/j.jpowsour.2015.02.142.
- [12] I.-Y. L. Hsieh, M. S. Pan, Y.-M. Chiang, and W. H. Green, "Learning only buys you so much: Practical limits on battery price reduction," *Appl. En*ergy, vol. 239, pp. 218–224, 2019, doi: 10.1016/j.apenergy.2019.01.138.
- [13] A. Sakti, I. M. Azevedo, E. R. Fuchs, J. J. Michalek, K. G. Gallagher, and J. F. Whitacre, "Consistency and robustness of forecasting for emerging technologies: The case of Li-ion batteries for electric vehicles," *Energy Policy*, vol. 106, no. 4, pp. 415–426, 2017, doi: 10.1016/j.enpol.2017.03.063.
- [14] H. Chen, J. C. Ho, and D. F. Kocaoglu, "A strategic technology planning framework: A case of Taiwan's semiconductor foundry industry," *IEEE Trans. Eng. Manage.*, vol. 56, no. 1, pp. 4–15, Feb. 2009, doi: 10.1109/TEM.2008.927815.
- [15] S. Kheybari, F. M. Rezaie, and J. Rezaei, "Measuring the importance of decision-making criteria in biofuel production technology selection," *IEEE Trans. Eng. Manage.*, vol. 68, no. 2, pp. 483–497, Apr. 2021, doi: 10.1109/TEM.2019.2908037.
- [16] C. Classi, D. Nowicki, J. Dickens, and A. Glassburner, "A redesign decision model for large-scale complex sustainmentdominated systems," *IEEE Trans. Eng. Manage.*, to be published, doi: 10.1109/TEM.2021.3092962.
- [17] L. Mauler, F. Duffner, W. G. Zeier, and J. Leker, "Battery cost forecasting: A review of methods and results with an outlook to 2050," *Energy Environ. Sci.*, vol. 14, no. 9, pp. 4712–4739, 2021, doi: 10.1039/D1EE01530C.
- [18] B. Nykvist, F. Sprei, and M. Nilsson, "Assessing the progress toward lower priced long range battery electric vehicles," *Energy Policy*, vol. 124, pp. 144–155, 2019, doi: 10.1016/j.enpol.2018.09.035.
- [19] R. E. Ciez and J. F. Whitacre, "Comparison between cylindrical and prismatic lithium-ion cell costs using a process based cost model," *J. Power Sources*, vol. 340, pp. 273–281, 2017, doi: 10.1016/j.jpowsour.2016.11.054.
- [20] G. Berckmans, M. Messagie, J. Smekens, N. Omar, and L. Vanhaverbeke, "Cost projection of state of the art lithium-ion batteries for electric vehicles up to 2030," *Energies*, vol. 10, no. 9, 2017, Art. no. 1314, doi: 10.3390/en10091314.
- [21] M. Wentker, M. Greenwood, and J. Leker, "A bottom-up approach to lithium-ion battery cost modeling with a focus on cathode active materials," *Energies*, vol. 12, no. 3, p. 504, 2019, doi: 10.3390/en12030504.
- [22] A. Sakti, J. J. Michalek, E. R. Fuchs, and J. F. Whitacre, "A technoeconomic analysis and optimization of Li-ion batteries for light-duty passenger vehicle electrification," *J. Power Sources*, vol. 273, no. 9, pp. 966–980, 2015, doi: 10.1016/j.jpowsour.2014.09.078.

- [23] F. Duffner, O. Krätzig, and J. Leker, "Battery plant location considering the balance between knowledge and cost: A comparative study of the EU-28 countries," *J. Cleaner Prod.*, vol. 264, 2020, Art. no. 121428, doi: 10.1016/j.jclepro.2020.121428.
- [24] M. Philippot, G. Alvarez, E. Ayerbe, J. van Mierlo, and M. Messagie, "Eco-efficiency of a lithium-ion battery for electric vehicles: Influence of manufacturing country and commodity prices on GHG emissions and costs," *Batteries*, vol. 5, no. 1, p. 23, 2019, doi: 10.3390/ batteries5010023.
- [25] R. J. Brodd and C. Helou, "Cost comparison of producing highperformance Li-ion batteries in the U.S. and in China," *J. Power Sources*, vol. 231, pp. 293–300, 2013, doi: 10.1016/j.jpowsour.2012.12.048.
- [26] H. Löfsten, "Industrialization of hybrid electric vehicle technology: Identifying critical resource dimensions," *J. Technol. Transfer*, vol. 41, no. 2, pp. 349–367, 2016, doi: 10.1007/s10961-015-9395-x.
- [27] X. Gu, P. Ieromonachou, L. Zhou, and M.-L. Tseng, "Optimising quantity of manufacturing and remanufacturing in an electric vehicle battery closed-loop supply chain," *Int. Mater. Data Syst.*, vol. 118, no. 1, pp. 283–302, 2018, doi: 10.1108/IMDS-04-2017-0132.
- [28] C. Scheller, K. Schmidt, and T. S. Spengler, "Decentralized master production and recycling scheduling of lithium-ion batteries: A technoeconomic optimization model," *J. Bus. Econ.*, vol. 91, no. 2, pp. 253–282, 2021, doi: 10.1007/s11573-020-00999-7.
- [29] M. Thomitzek, N. von Drachenfels, F. Cerdas, C. Herrmann, and S. Thiede, "Simulation-based assessment of the energy demand in battery cell manufacturing," *Procedia Corporate Insolvency Resolution Process*, vol. 80, no. 1, pp. 126–131, 2019, doi: 10.1016/j.procir.2019.01.097.
- [30] J. Schnell et al., "Data mining in lithium-ion battery cell production," J. Power Sources, vol. 413, pp. 360–366, 2019, doi: 10.1016/j.jpowsour.2018.12.062.
- [31] S. Thiede, A. Turetskyy, A. Kwade, S. Kara, and C. Herrmann, "Data mining in battery production chains towards multi-criterial quality prediction," *Corporate Insolvency Resolution Process Ann.*, vol. 68, no. 1, pp. 463–466, 2019, doi: 10.1016/j.cirp.2019.04.066.
- [32] S. Thiede, A. Turetskyy, T. Loellhoeffel, A. Kwade, S. Kara, and C. Herrmann, "Machine learning approach for systematic analysis of energy efficiency potentials in manufacturing processes: A case of battery production," *Corporate Insolvency Resolution Process Ann.*, vol. 69, no. 1, pp. 21–24, 2020, doi: 10.1016/j.cirp.2020.04.090.
- [33] A. Turetskyy, S. Thiede, M. Thomitzek, N. von Drachenfels, T. Pape, and C. Herrmann, "Toward data-driven applications in lithium-ion battery cell manufacturing," *Energy Technol.*, vol. 8, no. 2, 2020, Art. no. 1900136, doi: 10.1002/ente.201900136.
- [34] A. Turetskyy, J. Wessel, C. Herrmann, and S. Thiede, "Data-driven cyber-physical system for quality gates in lithium-ion battery cell manufacturing," *Procedia Corporate Insolvency Resolution Process*, vol. 93, no. 4, pp. 168–173, 2020, doi: 10.1016/j.procir.2020.03.077.
- [35] D. L. Wood, J. Li, and C. Daniel, "Prospects for reducing the processing cost of lithium ion batteries," *J. Power Sources*, vol. 275, pp. 234–242, 2015, doi: 10.1016/j.jpowsour.2014.11.019.
- [36] D. L. Wood, J. D. Quass, J. Li, S. Ahmed, D. Ventola, and C. Daniel, "Technical and economic analysis of solvent-based lithium-ion electrode drying with water and NMP," *Drying Technol.*, vol. 36, no. 2, pp. 234–244, 2018, doi: 10.1080/07373937.2017.1319855.
- [37] B. Ludwig et al., "Understanding interfacial-energy-driven dry powder mixing for solvent-free additive manufacturing of Li-ion battery electrodes," *Adv. Mater. Interfaces*, vol. 4, no. 21, 2017, Art. no. 1700570, doi: 10.1002/admi.201700570.
- [38] L. Wang et al., "Identifying the components of the solid-electrolyte interphase in Li-ion batteries," *Nature Chem.*, vol. 11, no. 9, pp. 789–796, 2019, doi: 10.1038/s41557-019-0304-z.
- [39] M. Winter, "The solid electrolyte interphase the most important and the least understood solid electrolyte in rechargeable Li batteries," *Zeitschrift für Physikalische Chemie*, vol. 223, no. 10/11, pp. 1395–1406, 2009, doi: 10.1524/zpch.2009.6086.
- [40] S. J. An, J. Li, Z. Du, C. Daniel, and D. L. Wood, "Fast formation cycling for lithium ion batteries," *J. Power Sources*, vol. 342, pp. 846–852, 2017, doi: 10.1016/j.jpowsour.2017.01.011.
- [41] H. H. Heimes et al., "The effects of mechanical and thermal loads during lithium-ion pouch cell formation and their impacts on process time," *Energy Technol.*, vol. 8, no. 2, 2020, Art. no. 1900118, doi: 10.1002/ente.201900118.
- [42] D. L. Wood, J. Li, and S. J. An, "Formation challenges of lithium-ion battery manufacturing," *Joule*, vol. 3, no. 12, pp. 2884–2888, 2019, doi: 10.1016/j.joule.2019.11.002.

- [43] D. L. Wood et al., "Perspectives on the relationship between materials chemistry and roll-to-roll electrode manufacturing for highenergy lithium-ion batteries," *Energy Storage Mater.*, vol. 29, no. 15, pp. 254–265, 2020, doi: 10.1016/j.ensm.2020.04.036.
- [44] J. P. McKinley, S. A. Sellers, and K. R. Colclazier, "Battery formation and charging system and method," U.S. Patent 2010164437, Jul. 1, 2010.
- [45] A. R. Jiménez et al., "A step towards understanding the beneficial influence of a LIPON-based artificial SEI on silicon thin film anodes in lithium-ion batteries," *Nanoscale*, vol. 10, no. 4, pp. 2128–2137, 2018, doi: 10.1039/C7NR06568J.
- [46] R. G. Fedorov, S. Maletti, C. Heubner, A. Michaelis, and Y. Ein-Eli, "Molecular engineering approaches to fabricate artificial solidelectrolyte interphases on anodes for Li-ion batteries: A critical review," *Adv. Energy Mater.*, vol. 11, no. 26, 2021, Art. no. 2101173, doi: 10.1002/aenm.202101173.
- [47] H. Wang and Y. Tang, "Artificial solid electrolyte interphase acting as 'armor' to protect the anode materials for high-performance lithiumion battery," *Chem. Res. Chin. Univ.*, vol. 36, no. 3, pp. 402–409, 2020, doi: 10.1007/s40242-020-0091-5.
- [48] Y. Liu, R. Zhang, J. Wang, and Y. Wang, "Current and future lithium-ion battery manufacturing," *iScience*, vol. 24, no. 4, 2021, Art. no. 102332, doi: 10.1016/j.isci.2021.102332.
- [49] C. Mao et al., "Balancing formation time and electrochemical performance of high energy lithium-ion batteries," *J. Power Sources*, vol. 402, pp. 107–115, 2018, doi: 10.1016/j.jpowsour.2018.09.019.
- [50] J. Wen, Y. Yu, and C. Chen, "A review on lithium-ion batteries safety issues: Existing problems and possible solutions," *Mater. Exp.*, vol. 2, no. 3, pp. 197–212, 2012, doi: 10.1166/mex.2012.1075.
- [51] N. Majid, S. Hafiz, S. Arianto, R. Y. Yuono, E. T. Astuti, and B. Prihandoko, "Analysis of effective pulse current charging method for lithium ion battery," *J. Phys., Conf. Ser.*, vol. 817, 2017, Art. no. 12008, doi: 10.1088/1742-6596/817/1/012008.
- [52] S. Bhattacharya and A. T. Alpas, "Micromechanisms of solid electrolyte interphase formation on electrochemically cycled graphite electrodes in lithium-ion cells," *Carbon*, vol. 50, no. 15, pp. 5359–5371, 2012, doi: 10.1016/j.carbon.2012.07.009.
- [53] K. A. Severson et al., "Data-driven prediction of battery cycle life before capacity degradation," *Nature Energy*, vol. 4, no. 5, pp. 383–391, 2019, doi: 10.1038/s41560-019-0356-8.
- [54] W. Diao, S. Saxena, and M. Pecht, "Accelerated cycle life testing and capacity degradation modeling of LiCoO₂-graphite cells," *J. Power Sources*, vol. 435, 2019, Art. no. 226830, doi: 10.1016/j.jpowsour.2019.226830.
- [55] A. Weng et al., "Predicting the impact of formation protocols on battery lifetime immediately after manufacturing," *Joule*, vol. 5, no. 11, pp. 2971–2992, 2021, doi: 10.1016/j.joule.2021.09.015.
- [56] J. Schnell, H. Knörzer, A. J. Imbsweiler, and G. Reinhart, "Solid versus liquid—a bottom-up calculation model to analyze the manufacturing cost of future high-energy batteries," *Energy Technol.*, vol. 8, no. 3, 2020, Art. no. 1901237, doi: 10.1002/ente.201901237.
- [57] J. Schnell, F. Tietz, C. Singer, A. Hofer, N. Billot, and G. Reinhart, "Prospects of production technologies and manufacturing costs of oxide-based all-solid-state lithium batteries," *Energy Environ. Sci.*, vol. 12, no. 6, pp. 1818–1833, 2019, doi: 10.1039/C8EE02692K.
- [58] L. Helmers et al., "Sustainable solvent-free production and resulting performance of polymer electrolyte-based all-solid-state battery electrodes," *Energy Technol.*, vol. 9, no. 3, 2021, Art. no. 2000923, doi: 10.1002/ente.202000923.
- [59] J. Sturm, A. Frank, A. Rheinfeld, S. V. Erhard, and A. Jossen, "Impact of electrode and cell design on fast charging capabilities of cylindrical lithium-ion batteries," *J. Electrochem. Soc.*, vol. 167, no. 13, 2020, Art. no. 130505, doi: 10.1149/1945-7111/abb40c.
- [60] K. Tsuruta, M. E. Dermer, and R. Dhiman, "Cell with a tabless electrode," U.S. Patent 11 588 184, Feb. 21, 2023.
- [61] K. G. Gallagher and P. A. Nelson, "Manufacturing costs of batteries for electric vehicles," in *Lithium-Ion Batteries*, Amsterdam, The Netherlands: Elsevier, 2014, pp. 97–126.
- [62] P. Nelson, D. Santini, and J. Barnes, "Factors determining the manufacturing costs of lithium-ion batteries for PHEVs," *World Elect. Veh. J.*, vol. 3, no. 3, pp. 457–468, 2009, doi: 10.3390/wevj3030457.
- [63] J. H. Schünemann, "Modell zur bewertung der herstellkosten von lithiumionenbatteriezellen," Braunschweig, 2015. [Online]. Available: https://www.researchgate.net/publication/322224024_Modell_zur_ Bewertung_der_Herstellkosten_von_Lithiumionenbatteriezellen

- [64] F. Duffner, M. Wentker, M. Greenwood, and J. Leker, "Battery cost modeling: A review and directions for future research," *Renewable Sustain. Energy Rev.*, vol. 127, 2020, Art. no. 109872, doi: 10.1016/j.rser.2020.109872.
- [65] A. Stephan, T. S. Schmidt, C. R. Bening, and V. H. Hoffmann, "The sectoral configuration of technological innovation systems: Patterns of knowledge development and diffusion in the lithium-ion battery technology in Japan," *Res. Policy*, vol. 46, no. 4, pp. 709–723, 2017, doi: 10.1016/j.respol.2017.01.009.
- [66] A. Malhotra, T. S. Schmidt, and J. Huenteler, "The role of inter-sectoral learning in knowledge development and diffusion: Case studies on three clean energy technologies," *Technol. Forecasting Social Change*, vol. 146, no. 1029, pp. 464–487, 2019, doi: 10.1016/j.techfore.2019.04.018.
- [67] A. Malhotra, H. Zhang, M. Beuse, and T. Schmidt, "How do new use environments influence a technology's knowledge trajectory? A patent citation network analysis of lithium-ion battery technology," *Res. Policy*, vol. 50, no. 9, 2021, Art. no. 104318, doi: 10.1016/j.respol.2021.104318.
- [68] B. A. Vojak and C. A. Suarez-Nunez, "Product attribute bullwhip in the technology planning process and a methodology to reduce it," *IEEE Trans. Eng. Manage.*, vol. 51, no. 3, pp. 288–299, Aug. 2004, doi: 10.1109/TEM.2004.830846.
- [69] O. Krätzig and N. Sick, "Exploring the role of entrepreneurial passion for facilitating university technology commercialization: Insights from battery research as an interdisciplinary field," *J. Eng. Technol. Manage.*, vol. 60, no. 86–94, 2021, Art. no. 101627, doi: 10.1016/j.jengtecman.2021.101627.
- [70] X. I. Quan, M. Loon, and J. Sanderson, "Innovation in the local context: A case study of BYD in China," *Int. J. Innov. Technol. Manage.*, vol. 15, no. 02, 2018, Art. no. 1850017, doi: 10.1142/S0219877018500177.
- [71] A. Stephan, C. R. Bening, T. S. Schmidt, M. Schwarz, and V. H. Hoffmann, "The role of inter-sectoral knowledge spillovers in technological innovations: The case of lithium-ion batteries," *Technol. Forecasting Social Change*, vol. 148, 2019, Art. no. 119718, doi: 10.1016/j.techfore.2019.119718.
- [72] J. Eitner, "Standortentscheidung ist gefallen: »forschungsfertigung batteriezelle« entsteht in münster," Munich, Germany, 2019. Accessed: Jan. 20, 2021. [Online]. Available: https://www.fraunhofer.de/de/ presse/presseinformationen/2019/juni/standortentscheidung-fuerforschungsfertigung-batteriezelle-gefallen.html
- [73] H. Yang and H. Jin, "SK innovation to start building second EV battery plant in United States," Accessed: Jan. 20, 2021. [Online]. Available: https://www.reuters.com/article/us-sk-innovationelectric/sk-innovation-to-start-building-second-ev-battery-plant-inunited-states-idUSKCN22B0IJ
- [74] S. Schaal, "Northvolt will produktionskapazität drastisch erhöhen," Accessed: Jan. 20, 2021. [Online]. Available: https://www.electrive.net/ 2019/11/26/northvolt-will-produktionskapazitaet-drastisch-erhoehen/
- [75] C. Hampel, "CATL starts building battery plant in Germany," Accessed: Jan. 20, 2021. [Online]. Available: https://www.electrive.com/2019/10/ 19/catl-starts-building-battery-plant-in-Germany/
- [76] F. Degen and M. Schütte, "Life cycle assessment of the energy consumption and GHG emissions of state-of-the-art automotive battery cell production," *J. Cleaner Prod.*, vol. 47, 2021, Art. no. 129798, doi: 10.1016/j.jclepro.2021.129798.
- [77] C. Hueber, K. Horejsi, and R. Schledjewski, "Review of cost estimation: Methods and models for aerospace composite manufacturing," *Adv. Manuf., Polym. Composites Sci.*, vol. 2, no. 1, pp. 1–13, 2016, doi: 10.1080/20550340.2016.1154642.
- [78] P. Duverlie and J. M. Castelain, "Cost estimation during design step: Parametric method versus case based reasoning method," *Int. J. Adv. Manuf. Technol.*, vol. 15, no. 12, pp. 895–906, 1999, doi: 10.1007/s001700050147.
- [79] F. Field, R. Kirchain, and R. Roth, "Process cost modeling: Strategic engineering and economic evaluation of materials technologies," *J. Met.*, vol. 59, no. 10, pp. 21–32, 2007, doi: 10.1007/s11837-007-0126-0.
- [80] K. Tornberg, M. Jämsen, and J. Paranko, "Activity-based costing and process modeling for cost-conscious product design: A case study in a manufacturing company," *Int. J. Prod. Econ.*, vol. 79, no. 1, pp. 75–82, 2002, doi: 10.1016/S0925-5273(00)00179-1.
- [81] Destatis, "Preise: Daten zur energiepreisentwicklung," Lange Reihen von Januar 2005 bis November 2020, Statistisches Bundesamt (Destatis), 2020. Accessed: Jan. 21, 2021. [Online]. Available: https://www.destatis.de/DE/Themen/Wirtschaft/Preise/Publikationen/ Energiepreise/energiepreisentwicklung-pdf-5619001.pdf;jsessionid= B2A7CF837E9FAE4AAAC4D17BC64C620F.internet8721?_blob= publicationFile

- [82] Z. Y. Zhang, "China electricity prices for industrial consumers," Accessed: Jan. 21, 2021. [Online]. Available: https://www.China-briefing. com/news/China-electricity-prices-industrial-consumers/
- [83] K. Lichtblau, C. Bähr, M. Krause, and T. Lang, "Wettbewerber der deutschen M+E-industrie: Gutachten im auftrag des arbeitgeberverbands gesamtmetall," Köln, Berlin, 2019. Accessed: Jan. 21, 2021. [Online]. Available: https://www.vme-net.de/de/system/files/downloads_ und_vorschaubilder/2019-06-03_gutachten_iw-consult.pdf
- [84] P. A. Nelson, S. Ahmed, K. G. Gallagher, and D. W. Dees, "Modeling the performance and cost of lithium-ion batteries for electric-drive vehicles," Third Edition, U.S. Department of Energy, Office of Scientific and Technical Information, 2019. Accessed: Jan. 21, 2021. [Online]. Available: https://www.osti.gov/biblio/1503280-modeling-performance-costlithium-ion-batteries-electric-drive-vehicles-third-edition
- [85] A. Eller, "Guidehouse: Lithium battery cell prices to almost halve by 2029, Accessed: Jan. 21 2021. [Online]. Available: https://www.energy-storage.news/news/guidehouse-lithium-batterycell-prices-to-almost-halve-by-2029
- [86] D. Chung, E. Elgqvist, and S. Santhanagopalan, "Automotive lithium-ion cell manufacturing: Regional cost structures and supply chain considerations," U.S. Department of Energy, 2016. Accessed: Jan. 21, 2021. [Online]. Available: https://www.nrel.gov/docs/fy16osti/66086.pdf
- [87] D. Küpper, K. Kuhlmann, S. Wolf, C. Pieper, G. Xu, and J. Ahmad, *The Future of Battery Production for Electric Vehicles*. Boston, MA, USA: Boston Consulting Group, 2018.
- [88] B3 report, "Chapter 11, LIB materials market bulletin (19Q1)," B3 Corporation, 2019.
- [89] J. Venkatasamy, "Lithium-ion batteries price trend and cost structure," Accessed: Jan. 21, 2021. [Online]. Available: https://www.beroeinc.com/ article/lithium-ion-batteries-price-trend-cost-structure/
- [90] N. Muralidharan et al., "Next-generation cobalt-free cathodes A prospective solution to the battery industry's cobalt problem," *Adv. Energy Mater.*, vol. 12, no. 9, 2022, Art. no. 2103050, doi: 10.1002/aenm.202103050.
- [91] M. Li et al., "SiO2/Cu/polyacrylonitrile-C composite as anode material in lithium ion batteries," *J. Power Sources*, vol. 240, pp. 659–666, 2013, doi: 10.1016/j.jpowsour.2013.05.017.
- [92] D. Wang, M. Gao, H. Pan, J. Wang, and Y. Liu, "High performance amorphous-Si@SiO /C composite anode materials for Li-ion batteries derived from ball-milling and in situ carbonization," *J. Power Sources*, vol. 256, pp. 190–199, 2014, doi: 10.1016/j.jpowsour.2013.12.128.
- [93] Z. Liu et al., "Silicon oxides: A promising family of anode materials for lithium-ion batteries," *Chem. Soc. Rev.*, vol. 48, no. 1, pp. 285–309, 2019, doi: 10.1039/C8CS00441B.
- [94] W. Zhao, G. Luo, and C.-Y. Wang, "Effect of tab design on large-format Li-ion cell performance," *J. Power Sources*, vol. 257, pp. 70–79, 2014, doi: 10.1016/j.jpowsour.2013.12.146.
- [95] D. Ligeois, P. Lacoste, and F. Fargeas, "Electric coupling of a connection to an electrochemical bundle," US20050008933A1, Jan. 13, 2015.
- [96] F. Degen and O. Kratzig, "Future in battery production: An extensive benchmarking of novel production technologies as guidance for decision making in engineering," *IEEE Trans. Eng. Manage.*, to be published, doi: 10.1109/TEM.2022.3144882.
- [97] S. Tschöcke et al., "Verfahren zum herstellen eines trockenfilms sowie trockenfilm und mit dem trockenfilm beschichtetes substrat," DE102017208220A1, Nov. 22, 2018.
- [98] J. Seeba, S. Reuber, C. Heubner, A. Müller-Köhn, M. Wolter, and A. Michaelis, "Extrusion-based fabrication of electrodes for high-energy Li-ion batteries," *Chem. Eng. J.*, vol. 402, 2020, Art. no. 125551, doi: 10.1016/j.cej.2020.125551.
- [99] M. Cudazzo, A. Lenz, and I. Landwehr, "Elektrostatische fluidisierungsvorrichtung und verfahren zur pulverbeschichtung eines substrats," DE102016212610A1, Feb. 15, 2018.
- [100] M. Gockeln, J. Glenneberg, M. Busse, S. Pokhrel, L. M\u00e4dler, and R. Kun, "Flame aerosol deposited Li4Ti5O12 layers for flexible, thin film all-solid-state Li-ion batteries," *Nano Energy*, vol. 49, pp. 564–573, 2018, doi: 10.1016/j.nanoen.2018.05.007.
- [101] The Federal Government Germany, "Foundations in place for CO2 pricing: Cabinet takes decision on national emissions trading," Accessed: Jan. 21, 2021. [Online]. Available: https://www.bundesregierung.de/ breg-en/issues/nationaler-emissionshandel-1685054
- [102] N. A.-Z. R-Smith, G. Gramse, M. Moertelmaier, M. Kasper, and F. Kienberger, "Advanced self-discharge measurements of lithium-ion cells and comparison to modeling," in *Proc. IEEE Int. Instrum. Meas. Technol. Conf.*, 2020, pp. 1–5.

- [103] E. Ayerbe, M. Berecibar, S. Clark, A. A. Franco, and J. Ruhland, "Digitalization of battery manufacturing: Current status, challenges, and opportunities," *Adv. Energy Mater.*, vol. 12, no. 17, 2022, Art. no. 2102696, doi: 10.1002/aenm.202102696.
- [104] C. Schneider, M. Rosmann, A. Losch, and A. Grunwald, "Transformative vision assessment and 3-D printing futures: A new approach of technology assessment to address grand societal challenges," *IEEE Trans. Eng. Manage.*, vol. 70, no. 3, pp. 1089–1098, Mar. 2023, doi: 10.1109/TEM.2021.3129834.
- [105] V. L. Mitchell and R. W. Zmud, "Endogenous adaptation: The effects of technology position and planning mode on IT-enabled change," *Decis. Sci.*, vol. 37, no. 3, pp. 325–355, 2006, doi: 10.1111/j.1540-5414.2006.00130.x.
- [106] C. Thies, K. Kieckhäfer, and T. S. Spengler, "Activity analysis based modeling of global supply chains for sustainability assessment," J. Bus. Econ., vol. 91, no. 2, pp. 215–252, 2021, doi: 10.1007/s11573-020-01004-x.



Florian Degen received the Dipl.-Ing. degree in mechanical engineering with a focus on production and manufacturing technology from RWTH Aachen University, Aachen, Germany, in 2011. He received the doctorate degree in technology development of novel manufacturing processes, in 2015, and completed further studies as "Executive Master in Business Administration" at RWTH Business School and HSG St. Gallen, St. Gallen, Switzerland in the field of "Technology and Innovation Management," in 2018. In addition, during his studies, for over a year,

he was a Visiting Engineer with McMaster Manufacturing Research Institute, Hamilton, Canada and with the Singapore Institute of Manufacturing Technology, Singapore. From 2011 to 2019, he held various positions with the Fraunhofer Institute for Production Technology, Aachen, Germany, most recently as Department Head responsible for the technology development of novel manufacturing processes. He is currently a Division Manager with the Strategy and Corporate Development, Fraunhofer Research Institution for Battery Cell Production FFB, Münster, Germany. In this function, he is responsible for the abovementioned topics as well as for technology management, sustainability management and business development.



Oliver Krätzig received the B.Sc. degrees in industrial engineering and business and engineering from the University of Applied Sciences Münster, Münster, Germany, in 2013, the M.Sc. degree in business and engineering from the Clausthal University of Technology, Clausthal-Zellerfeld, Germany, in 2016, and the Ph.D. "Dr. rer. nat." degree in business chemistry from the University of Münster, Münster, in 2019

Since 2016, he has been working on technology transfer issues from battery research in his position as Research Assistant with Helmholtz Institute Münster,

Münster, Germany. After graduating, he entered the Fraunhofer Society as Research Fellow and Technology Manager, first with the Institute for Systems and Innovation Research ISI, Karlsruhe, Germany, and then with the Institute for Production Technology IPT, Aachen, Germany. After the transformation of the IPT-division "Research Institution for Battery Cell Production FFB" to an independent Fraunhofer institution in 2022, he continued his activities as Technology Manager and his research work in the field of technology assessment and corresponding method development for battery cell production. His research interests include issues of technology and innovation management, such as technology transfer for the targeted promotion of energy storage technologies.

Dr. rer. nat. Krätzig has been a Member of the German Chemical Society since 2017.